Author's response to reviews

Title: High prevalence of tuberculosis among university students in Northwest Ethiopia: A retrospective study

Authors:

- Beyene Moges (beyemoges@gmail.com)
- Bemnet A Tedla (amarebem6@gmail.com)
- Gizachew Yismaw (gizachewyismaw@yahoo.com)
- Meseret Workineh (mwmesi@gmail.com)
- Shitaye Alemu (shitayea@yahoo.com)
- Desalew Mekonnen (desalewm@yahoo.com)
- Ermias D Ejjara (ermi_diro@yahoo.com)
- Belay Tesema (bt1488@yahoo.com)
- Afework K Gizaw (afeworkkassu@yahoo.com)

Version: 2
Date: 26 December 2013

Author's response to reviews:

I have added an "Acknowledgement" section with the following content:

The study was financially supported by the University of Gondar, Ethiopia. We would like to thank the staff of the DOTS clinic at University of Gondar Hospital without whom this study could not have been completed.